Ther Adv Gastroenterol

2023, Vol. 16: 1–14

17562848231183529

© The Author(s), 2023. Article reuse guidelines: sagepub.com/journalspermissions

Tamás Resál+[®], Péter Bacsur+[®], Miklós Horváth+, Kata Szántó[®], Mariann Rutka, Anita Bálint, Anna Fábián[®], Renáta Bor[®], Zoltán Szepes, János Fekete, Klaudia Farkas*, Pál Miheller* and Tamás Molnár*

Nationwide experiences with trough levels,

durability, and disease activity among

inflammatory bowel disease patients

following COVID-19 vaccination

Abstract

Background: The severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) pandemic has complicated the management of inflammatory bowel diseases (IBD).

Objectives: This study aimed to assess the efficacy of different anti-SARS-CoV-2 vaccines under different treatments in IBD patients and identify predictive factors associated with lower serological response, including anti-tumor necrosis factor (anti-TNF) drug levels. **Design:** A prospective, double-center study of IBD patients was conducted following messenger ribonucleotide acid (mRNA) and non-mRNA anti-SARS-CoV-2 vaccination.

Methods: Healthy control (HC) patients were enrolled to reduce bias. Baseline and control samples were obtained 14 days after the second dose to assess the impact of conventional and biological treatments. Clinical and biochemical activity, serological response level, and anti-TNF drug levels were measured.

Results: This study included 199 IBD (mean age, 40.9 ± 12.72 years) and 77 HC participants (mean age, 50.3 ± 12.36 years). Most patients (76.9%) and all HCs received mRNA vaccines. Half of the IBD patients were on biological treatment (anti-TNF 68.7%). Biological and thiopurine combined immunomodulation and biological treatment were associated with lower serological response (p < 0.001), and mRNA vaccination promoted better antibody levels (p < 0.001). Higher adalimumab levels caused lower serological response (p = 0.006). W8 persistence of anti-SARS-CoV-2 level was equal in IBD and HC groups. Vaccination did not aggravate clinical disease activity (p = 0.65).

Conclusion: Anti-SARS-CoV-2 vaccination is considerably efficacious in IBD patients, with mRNA vaccines promoting better antibody levels. The negative impact of combined biological treatment, especially with high adalimumab drug levels, on serological response to vaccination should be considered. Although midterm durability of vaccination is encouraging, more data are needed to expand the existing understanding on this issue.

Plain language summary

Adjustment of COVID-19 vaccination to adalimumab trough level is considerable due to the reduced serological response. mRNA vaccination should be preferred in case of IBD patients with an equal durability of anti-SARS-CoV-2 level of subjects and healthy control participants.

Keywords: anti-TNF, COVID-19, durability, IBD, persistence, vaccination

Received: 27 January 2023; revised manuscript accepted: 5 June 2023.

Correspondence to: Tamás Molnár

Department of Medicine, Albert Szent-Györgyi Medical School, University of Szeged, Kálvária Avenue 57, Szeged, H-6720, Hungary.

molnar.tamas@med.uszeged.hu

Tamás Resál

Péter Bacsur Kata Szántó Mariann Rutka Anita Bálint Anna Fábián Renáta Bor Zoltán Szepes Klaudia Farkas Department of Medicine, Albert Szent-Györgyi Medical School, Universit

Medical School, University of Szeged, Szeged, Hungary

Miklós Horváth Pál Miheller

Department of Surgery and Interventional Gastroenterology, Faculty of Medicine, Semmelweis University, Budapest, Hungary

János Fekete

Department of Bioinformatics, Semmelweis University, Budapest, Hungary

*Authors share co-first authorship.

*Authors share senior authorship.

journals.sagepub.com/home/tag

CC O S BY NC

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the Sage and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

Introduction

The Coronavirus Disease 2019 (COVID-19) pandemic caused by severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) has raised issues in the management of inflammatory bowel diseases (IBD), namely ulcerative colitis (UC) and Crohn's disease (CD).¹ Concerns regarding the outcome of COVID-19 in relation to IBD or an immunocompromised state emerged first.^{2,3,4} A significant proportion of IBD patients are considered to be immunosuppressed. Patients receiving >20mg of corticosteroids, thiopurines, or biologics clearly belong to this group, although malnourished patients can also be considered to have a higher risk for infections.^{5,6} While clinical trials investigating the efficacy of vaccines against SARS-CoV-2 exclude immunosuppressed patients, all professional organizations recommend all types of COVID-19 vaccinations for patients with IBD.7,8,9 A population-based vaccination program involving the adenovirus vector vaccine (Sputnik[®], Gamaleya Research Institute of Epidemiology and Microbiology) and inactivated virus vaccine (Sinopharm[®], Sinopharm's Beijing Institute of Biological Products) started relatively early in Hungary, with messenger ribonucleotide acid (mRNA) vaccines following a little later. However, while most gastroenterologists advised patients to receive mRNA-based vaccination, a portion of IBD patients received a vector vaccine or inactivated virus vaccine at the early stages of the national vaccination program.¹⁰ There was some uncertainty regarding the efficacy of these vaccines among both patients and treating physicians, considering that the European Medical Agency had yet to approve their administration when they were first available.

Recently published meta-analyses have investigated the effectiveness, safety, and durability of anti-SARS-CoV-2 vaccination among IBD patients. Accordingly, Bhurwal et al. reported a cumulative seroconversion rate of 96.8% after the second vaccine dose.¹¹ Moreover, they found no significant differences in serological response according to the ongoing treatment. The same study, which collected the results of 21 articles mostly on mRNA- and vector-type vaccines, found an adverse event rate of 0.09%, most of which were mild. Jena et al. analyzed available literature regarding the durability of anti-SARS-CoV-2 vaccination in IBD patients in addition to the efficacy.¹² After synthesizing data from 46 studies with more than 9000 IBD patients, they found an overall seroconversion rate of 96% after complete vaccination based on around 31 eligible articles. Subgroup analysis according to vaccine type showed a 96–98% and 78–90% seroconversion rate in cases receiving mRNA and vector vaccines. Based on nine enrolled studies, they concluded that titers decrease 4 weeks after anti-SARS-CoV-2 vaccination.

Although the large body of clinical data published to date have answered many questions, to our knowledge, few data exist on the relationship between serum levels of biological therapy (BT) and the rate of seroconversion. Data regarding non-mRNA vaccinations among immunosuppressed IBD patients are lacking, with no direct comparison having been performed. Moreover, it remains unknown whether any predictors of immune response might influence vaccination strategy in this cohort. Patients are often hesitant regarding vaccine administration owing to fears, based on perceived or real information or experience, that the vaccine will result in disease flareups. Finally, given that only a few studies have used adequate controls, comparative data on the durability and persistence of responses between immunosuppressed patients and healthy controls (HCs) have been lacking.

Aims

We conducted a double-center, prospective cohort study to compare the level of the seroconversion after vaccination with different types of COVID-19 vaccines in IBD patients. Our primary objective was to investigate the serological response and persistence in the levels of specific anti-SARS-CoV-2 S antibodies after the administration of different SARS-CoV-2 vaccines in IBD patients receiving various therapeutic agents and compared them to those in HCs. Our secondary objective was to identify clinical and biochemical predictors associated with inadequate serological response. Furthermore, we aimed to assess the effects of vaccination on disease activity. Our tertiary objective was to investigate serum anti-TNF levels and serological response rates in IBD patients following the administration of different vaccines against SARS-CoV-2.

Materials and methods

Study design and settings

We conducted a double-center, prospective cohort study between March 2021 and February

2022. at the University of Szeged and Semmelweis University, Hungary. Patients who satisfied the inclusion/exclusion criteria were divided into four groups according to their treatment. A distinction was made between those receiving combined biologic and immunosuppressive therapy (COMB), those receiving biologic (BT) or immunosuppressant monotherapy [azathioprine (AZA)] and those who did not receive either biologic or immunosuppressive therapy (NONE). Eligible patients were requested to visit for blood sampling within 24 h after their first vaccination. Patients received the second dose of the vaccine at intervals specified in the vaccine's label. Serological testing of each subject was performed 14 days after their second vaccination. The reporting of this study conforms to the STROBE statement.13

Participants

Consecutive patients diagnosed with IBD who visited the outpatient department were enrolled in this study. Healthy subjects included in the H-UNCOVER randomized trial were used as the control group.¹⁴ Participants with a history of COVID infection and those <18 years of age were excluded: blood tests were performed at the scheduled clinical control, during which a sero-logical test was performed to rule out previous asymptomatic COVID infection. Participation was voluntary, and data were collected anonymously. Subjects consented to the use of data only for scientific purposes.

Data source and measurements

Demographic and clinical data were obtained at baseline. Sex, age at inclusion, type of IBD, ongoing treatment, disease classification (according to Montreal definitions¹⁵), and disease activity were recorded. Clinical activity was assessed using the Crohn's Disease Activity Index (CDAI¹⁶) and pMayo score (pMayo¹⁷) in UC patients. Biochemical activity was assessed using C-reactive protein (CRP) measurements. Extended disease was defined as bowel involvement proximal to the rectum in cases with UC and as at least two involved segments in cases with CD. The type of vaccine was registered, and serum infliximab or serum adalimumab concentrations were determined at this time point. The anti-SARS-CoV-2 S (S) antibody levels were measured using the Elecsys Anti-SARS-CoV-2 Spike Antibody Immunoassay[®] (Roche[®], Basel, Switzerland), with the cutoff value set at 0.8 U/mL according to the manufacturer's protocol. The assay had a sensitivity of > 99.5% for confirming SARS-CoV-2 infection on the 14th day following polymerase chain reaction (PCR) as per the product's label.

Serum infliximab (IFX; #Ridascreen IXF Monitoring[®], R-Biopharm[®], Darmstadt, Germany) and adalimumab (ADA; #Ridascreen ADM Monitoring[®], R-Biopharm[®], Darmstadt, Germany) concentrations were determined using the ELISA method as per the manufacturer's protocol (R-Biopharm[®], Darmstadt, Germany). The sensitivity of the IFX and ADA assays was <1 ng/ mL, respectively. The intra- and inter-assay coefficients of variation for both assays were <15%.

Additional data were obtained by conducting two sub-analyses at weeks 4 and 8 following the second anti-SARS-CoV-2 vaccination in one of our tertiary referral centers (University of Szeged). We assessed persistence of serological response and the effect of vaccination on disease activity during the follow-up period. Anti-SARS-CoV-2 S antibody levels were measured at baseline (before vaccination) and 4 and 8 weeks after the second vaccination. In this sub-analysis, clinical activity was interpreted using the Physician's Global Assessment (PGA) score.¹⁸ Clinical and biochemical activities were compared at baseline and again 8 weeks after the second dose.

Statistical analysis

Statistical analysis was performed via IBM SPSS software (IBM SPSS Statistics for Windows, Version 26.0, IBM Corp., Armonk, NY, USA). Normality was tested using visual interpretations. Descriptive statistics were interpreted as mean \pm standard deviation of the mean (SD) for continuous variables and count + percentages for categorical variables. After checking assumptions, the Welch test or Mann-Whitney test and Kruskal-Wallis test were applied to compare groups described with continuous variables. Significance values had been adjusted using the Bonferroni correction for multiple tests. On the other hand, groups described with categorical variables were compared using the chi-squared

Table 1.	Baseline	demograph	nic data	of IBD	patients
----------	----------	-----------	----------	--------	----------

Characteristics	IBD (<i>n</i> = 199)	HC (<i>n</i> = 77)
age, mean (±SD)	40.9 (±12.72)	50.3 (±12.36)
gender, male N (%)	95 (47.7)	21 (27.3)
vaccine type N (%)		
mRNA (%)	153 (76.9)	77 (100)
Pfizer	120 (78.4)	77 (100)
Moderna	33 (21.6)	0 (0)
non-mRNA (%)	46 (23.1)	0 (0)
AstraZeneca	23 (50.0)	0 (0)
Sputnik V	11 (23.9)	0 (0)
Janssen	1 (2.2)	0 (0)
Sinopharm	11 (23.9)	0 (0)

SD, standard deviation of mean; *N*, number of subjects; mRNA, messenger ribonucleotide acid.

test and Fisher's exact test. A p value of <0.05 indicated statistical significance. To reduce bias, propensity score matching (using age, sex, and type of vaccine as variables) was used to select HC patients.

To examine predictive factors associated with serological response, linear regression models were constructed using age, BT, vaccine type, disease type, concomitant corticosteroid treatment, disease duration, extended disease, and clinical and biochemical activities as variables. Linear regression models were constructed to assess the relationship between anti-TNF drug levels and serological response.

To measure serological persistence, the Welch test was used based on ln + 1 values of anti-SARS-CoV-2 S antibody levels.

Results

Baseline characteristics

This study included 199 IBD patients (male/ female ratio 95/104, mean age 40.9 ± 12.72 years). More patients had CD than UC (n=127, 63.8%*versus* n=72, 36.2%). Moreover, propensity score matching from a database including 105 patients was used to select 77 HCs. The HCs were older than IBD patients $(50.3 \pm 12.36 \text{ versus}$ $40.94 \pm 12.72 \text{ years}; p < 0.001$). Most of the patients received mRNA-type vaccines (n=153,76.9%), whereas 46 patients (23.1%) received non-mRNA vaccines. Most of the HC participants received mRNA-type vaccines (n=66,85.7%). Baseline demographic data and type of vaccinations are shown in Tables 1 and 2.

In total, 63.8% of the patients had CD, with most cases having ileocolonic localization and inflammatory phenotype (38.6% and 48.8%, respectively). Almost half of the UC patients had pancolitis (47.22%). Moreover, 49.7% of the patients were in the BT group, and more than two-thirds of them were on anti-TNF therapy (68.7%). In total, 11.6% of the patients received azathioprine as monotherapy (AZA group), 22.1% received it in combination with biological agents (COMB group), and 16.6% received neither biologics nor azathioprine (NONE group). Based on the clinical activity indexes, most of the CD patients were in clinical remission (mean CDAI 85.66 ± 58.8), whereas UC patients showed remission to mild disease activity (mean pMayo 1.27 ± 1.3 ; Table 2.).

Serological response to vaccination across different groups

Following all-type and mRNA vaccinations, anti-SARS-CoV-2 S antibody levels were significantly higher in the NONE group (p < 0.001); however, no significant difference between the groups was observed among cases receiving non-mRNA vaccination (p=0.447). Further details are available in Table 3.

Anti-SARS-CoV-2 S antibody titers in patients showed a decreasing trend in the following order of treatment: AZA, HC, BT, and combined biologic and immunosuppressive therapy (mean values of mRNA vaccination subgroup: NONE group: 8179 U/mL, AZA group: 4880 U/mL, HC group: 1931 U/mL, BT group: 1861 U/mL, COMB: 1624.5 U/mL; p < 0.001). Anti-SARS-CoV-2 S antibody levels were significantly higher in the NONE group compared to the BT group (p=0.003), COMB (p < 0.001), and HC (p < 0.001). Post hoc analysis was performed to determine the effect of vedolizumab (VDZ) and ustekinumab (UST) compared to the NONE group in serological response, which revealed no

Characteristics	IBD (n=199)
Disease type, CD N (%)	127 (63.8)
Disease duration, years, median (IQR)	12 (6–18)
Disease location*, N (%)	
lleum	37 (29.1)
Colon	40 (31.5)
lleocolic	49 (38.6)
Upper gastrointestinal involvement	4 (2.5)
Disease behavior*, N (%)	
Inflammatory disease	62 (48.8)
Stricturing disease	25 (19.7)
Penetrating disease	40 (31.5)
Age classification*, N (%)	
<16 years	9 (7.9)
17–39	82 (64.6)
40+	36 (28.3)
Disease extension*, N (%)	
E1 proctitis	9 (12.5)
E2 distal colitis	29 (40.28)
E3 pancolitis	34 (47.22)
Biological therapy group N (%)	99 (49.7)
Infliximab	36 (36.4)
Adalimumab	32 (32.3)
Vedolizumab	7 (7.1)
Ustekinumab	14 (14.1)
Tofacitinib	10 (10.10)
Azathioprine group N (%)	23 (11.6)
Combined group N (%)	44 (22.1)
None group N (%)	33 (16.6)

Table 2.	Baseline	clinical	data	of IBI	D patients.
----------	----------	----------	------	--------	-------------

Table 2. (Continued)

Characteristics	IBD (n=199)		
Disease activity mean (±SD)			
CDAI	85.66 (58.803)		
pMayo	1.27 (1.127)		
CRP	6.371 (13.336)		
CD, Crohn's disease; CDAI, Crohn's disease activity index; CRP, C-reactive protein; IQR, interquartile range; mRNA, messenger ribonucleotide acid; <i>N</i> , number of subjects, SD, standard deviation of mean. *Assessed by Montreal classification.			

significant differences (p=0.698). Analysis of TOFA *versus* NONE group was not performed due to the limiting effect of the size of TOFA group. Other comparisons did not show any significance. Table 3 and Figure 1 provide further data regarding the serological response to vaccination.

mRNA vaccination (p < 0.05) promoted better serological reponse compared to non-mRNA vaccination (p=0.571) in all cases except the vedolizumab treatment group.

According to our model, mRNA vaccines were associated with higher serological response (B = -0.523; p < 0.001). In addition, age had a negative impact on anti-SARS-CoV-2 S antibody levels (B=-0.169; p=0.014), and biological treatment was associated with lower serological response (B=-0.163; p=0.016). Clinical and biochemical (CRP and lymphocyte count) activities or disease type did not influence anti-SARS-CoV-2 S antibody levels according to the same model. Concomitant corticosteroid usage, disease duration, and disease extent had no significant impact on serological response (B=-0.130,p=0.074; B=-0.102, p=0.205; B=0.017,p=0.813). Coupling data are shown in Tables 3-4. Model details with selected variables are available in Table 5.

Serological response and anti-TNF serum level

(Continued)

80.0% of patients receiving ADA and 76.7% of the IFX group have been vaccinated with

THERAPEUTIC ADVANCES in

Table 3. Serological response in different groups.

Variables	BT group (<i>n</i> = 99)	AZA group (<i>n</i> =23)	COMB group (<i>n</i> =44)	NONE group (n = 33)	HC group (<i>n</i> = 77)	p Value
age, mean (±SD)	43.3 (11.80)	40.7 (15.12)	35.7 (9.43)	41.0 (15.63)	50.3 (12.36)	< 0.001
mRNA vaccine <i>N</i> (%)	73 (73.7)	17 (73.9)	34 (77.3)	29 (87.9)	66 (85.7)	0.214
non-mRNA vaccine <i>N</i> (%)	26 (26.3)	6 (26.1)	10 (22.7)	4 (12.1)	11 (14.3)	
anti-SARS-CoV-2 S level (U/ml)	1147 (302–4678)	3381 (251–7988)	976.5 (251–3937)	6122 (2334–13,808)	1629 (588–2815)	<0.001
mRNA anti- SARS-CoV-2 S level (U/ml)	1861 (666–6617)	4880 (2767–13,500)	1624.5 (384–4750)	8179 (2765–14,471)	1931 (868–2934)	<0.001
non-mRNA anti- SARS-CoV-2 S level (U/ml)	175 (38.4–1009)	73.3 (1.8–3354)	230.1 (39.8–533.5)	1562.8 (298.8–3400.5)	167 (125–358)	0.447

AZA, azathioprine; BT, biological therapy; COMB, biological therapy and azathioprine combination; HC, healthy control; mRNA, messenger ribonucleotide acid; n, number of subjects; SD, standard deviation of mean.



Figure 1. Comparison of anti-SARS-CoV-2 S antibody levels following mRNA vaccinations between groups. Anti-SARS-CoV-2 S antibody levels of the NONE group were significantly higher compared to the BT (p = 0.003), COMB (p < 0.001), and HC groups (p < 0.001). No significant difference was observed between the NONE and AZA groups (p = 0.99).

mRNA vaccination. No difference was observed in the distribution of patients between vaccinations groups ($\chi^2 = 0.12$; p = 0.73); therefore, we assessed the impact of the serum IFX and ADA levels on anti-SARS-CoV-2 S antibody titers. Accordingly, we found no significant correlation between serum IFX levels and serological response (B=0.332; p=0.078). However, higher ADA levels were associated with lower anti-SARS-CoV-2 S antibody levels (B=-0.404; p=0.006). Significant data are summarized in

Treatments	mRNA (n=219)	non-mRNA (n = 57)	p Value
All subjects	2540 (758–5822)	188 (40.4–772)	<0.001
BT group	1861 (666–6617)	175 (38.4–1009)	< 0.001
Infliximab (n=36)	1147 (386–3839)	198.5 (39.8–772.0)	< 0.001
Adalimumab (<i>n</i> =32)	1556 (523–4108)	209.1 (124.8–251.0)	< 0.001
Vedolizumab (<i>n</i> =7)	3207 (650.5–7764)	2167.5 (835.25-5332)	0.571
Ustekinumab (<i>n</i> = 14)	10328 (8359.5–20,488.5)	102.7 (22.84–3533)	0.005
Tofacitinib (<i>n</i> = 10)	1339.5 (747–3018)	113 (20.8–174)	< 0.001
AZA group (<i>n</i> = 23)	4880 (2767–13500)	73.30 (1.8–3354)	0.008
COMB group (<i>n</i> =44)	1624.5 (384–4750)	230.1 (39.8–533.5)	0.001
NONE group (<i>n</i> =33)	8179 (2765–14471)	1562.8 (298.8–3400.5)	0.027
HC group (<i>n</i> = 77)	1931 (868–2934)	167 (125–358)	< 0.001

Table 4. Anti-SARS-CoV-2 (S) levels according to the type of the vaccine across different treatments.

BT, biological therapy; AZA, azathioprine; COMB, biological therapy and azathioprine combination; HC, healthy control; n, number of subjects; SD, standard deviation of mean.

Table 5.	Linear regression	model to assess h	nigher serologica	l response. Mod	el summary: <i>R</i> = 0.627,	R2=0.393,
F = 12.779	9, and <i>p</i> < 0.001.					

Variables	В	t	p Value	95.0% CI for	.В
(Constant)		10.362	0.000	7.586	11.164
Age	-0.169	-2.497	0.014	-0.054	-0.006
Biological treatment (0: no, 1: yes)	-0.163	-2.442	0.016	-1.623	-0.171
Vaccine category (1: mRNA; 2:non-mRNA)	-0.523	-7.729	0.000	-3.493	-2.070
ln CRP	0.112	1.643	0.103	-0.049	0.530
Disease (UC and CD)	-0.041	-0.603	0.548	-0.843	0.449
Lymphocyte count	0.111	1.617	0.108	-0.004	0.042
Clinical activity	-0.089	-1.316	0.190	-1.214	0.244

B, standardized coefficients beta; CD, Crohn's disease; CI, Confidence interval; CRP, C-reactive protein; UC, ulcerative colitis.

Table 6. Figure 2 shows the connection between adalimumab drug level and serological response.

Persistence of SARS-CoV-2 S antibody levels following mRNA vaccination

Based on the results of our single center subanalysis, follow-up data of 100 participants were collected (IBD n=61, HC n=39) after mRNA vaccination. Age was statistically similar in both groups (p=0.53). No significant difference was observed between the IBD and HC groups either before the second dose (p=0.091) or at weeks 4 (p=0.084) and 8 (p=0.953) after the second dose of the vaccine. Coupling data are detailed in Table 7 and Figure 3.

Table 6. Linear regression model to assess serological response influence of ADA level. Model summary: n = 45, R = 0.404, $R^2 = 0.163$, F = 8.395, and p < 0.001.

Variable	В	t	p Value	95.0% CI for <i>B</i>	
(Constant)		8.162	0.000	6.754	11.196
adalimumab level	-0.404	-2.897	0.006	-1.664	-0.298
P standardized sofficients hets. CL confidence interval					

B, standardized coefficients beta; CI, confidence interval.



Figure 2. Linear regression model regarding ln logarythm of adalimumab trough level and serological response. Higher adalimumab level was associated with reduced serological response to anti-SARS-CoV-2 vaccination (B = -0.404, p = 0.006).

 Table 7.
 Anti-SARS-CoV-2 levels during follow-up among IBD and HC participants who received mRNA vaccines.

Variables	IBD mean (±SD) <i>n</i> =61	HC mean (±SD) <i>n</i> =39	p Value		
Age, years	47.2 (12.5)	48.6 (10.4)	0.534		
Before second dosage	1181.84 (2939)	75.16 (90)	0.091*		
After 4 weeks of second dosage	4114.79 (5515)	2278.35 (3090)	0.084*		
After 8 weeks of second dosage	2860.53 (5068)	1464.54 (943)	0.953*		
HC, healthy control; SD, standard deviation of mean.					

*Comparisons of groups based on ln + 1 values.

Impact of anti-SARS-CoV-2 vaccination on disease activity

Follow-up data for 81 and 66 IBD participants were analyzed at baseline and 8 weeks after the second dose of anti-SARS-CoV-2 vaccination. CRP levels, a marker of biochemical activity, significantly decreased from a mean baseline level of 5.65 ± 8.34 mg/L to a mean level of 4.02 ± 3.45 mg/L at week 8 after the second vaccine dose (p = 0.038). No significant difference in clinical disease activity was observed between baseline and follow-up measurements (0.43 ± 0.74 and 0.41 ± 0.61 ; p = 0.65). Related data are summarized in Table 8.



Figure 3. Persistence of anti-SARS-CoV-2 S antibody levels during follow-up period in IBD patients after mRNA vaccination. No significant difference was observed before (p = 0.091), after 4 weeks (p = 0.084), and after 8 weeks (p = 0.953) of the second vaccine dose.

Table 8. Change in clinica	and biochemical (CRP)	activity during follow-up
----------------------------	-----------------------	---------------------------

Disease activity	Baseline n=81	After 8 weeks of second dosage n=66	p Value	
CRP, mg/l, mean (±SD)	5.65 (8.34)	4.02 (3.45)	0.038	
PGA, n (%)				
inactive	56 (69.1)	43 (65.2)	0.65	
mild	17 (21.0)	19 (28.8)		
moderate	6 (7.4)	3 (4.5)		
severe	2 (2.5)	1 (1.5)		
CRP, C-reactive protein; PGA, patient global assessment score; SD, standard deviation of mean.				

Discussion

The current study focused on the serological response after SARS-CoV-2 immunization considering following issues. Contradictory and limited data have been available among immuno-compromised patients after anti-SARS-CoV-2 immunization. To our knowledge, our prospective cohort analysis has been the first unique study to compare different types of vaccines (mRNA and non-mRNA) and biological and/or immuno-suppressive treatment on serological response in a well-defined cohort.

Our double-center, prospective cohort study analyzed the data of 199 IBD and 77 HC participants.BasedonHungarianIBDrecommendations, most patients received mRNA-type anti-SARS-CoV-2 vaccines; however, to compare our findings to internationally existing data, we also analyzed the non-mRNA vaccines. In accordance with clinical practice guidelines, anti-TNF therapy was the most common biological treatment among our enrolled patients.^{19,20}

Our study confirmed that the mRNA vaccines promoted superior seroconversion levels than did non-mRNA vaccines among immunocompromised IBD patients. Comparing non-mRNA vaccinations, no difference was observed between attenuated adenovirus vaccines and inactivated whole virion vaccines, according to an Indian prospective cohort study.²¹ However, data are still lacking regarding non-mRNA vaccinations in IBD populations. Among IBD patients, ongoing biological, and/or immunomodulatory treatment definitely resulted in lower antibody response. Although higher ADA levels were associated with lower serological response, no significant difference was observed in cases receiving IFX. The durability of anti-SARS-CoV-2 S antibody levels did not differ between the IBD and HC groups 8 weeks after the administration of mRNA vaccines. Our results suggest no causal relationship between disease flares and immunization based on clinical and biochemical parameters.

Several studies have analyzed the possible effects of biologic treatments on seroconversion in IBD patients due to both infection and vaccination. A multicenter prospective observational cohort study in the United Kingdom (CLARITY) had shown that IFX significantly attenuated seroprevalence, seroconversion, and the magnitude of anti-SARS-CoV-2 antibody reactivity after SARS-CoV-2 infection, especially in the combination therapy group, compared to vedolizumab.²² A retrospective study in the same population showed that the negative effects of ADA on serological response were similar to those of IFX. In cases with undetectable TNF inhibitor levels, the seropositivity rate was comparable to vedolizumab.23

However, more data have emerged in the past few months. After recruiting consecutive IBD patients, the CLARITY study investigated the immunogenicity to BNT162b2 and ChAdOx1 nCoV-19. Unsurprisingly, IFX therapy promoted a lower serologic response not only after infection but also after a single vaccine dose, especially in those receiving combination treatment, compared to vedolizumab monotherapy. This effect was blunted after the second vaccine dose.²²

Another prospective multicenter study from Israel found that although two doses of BNT162b2 seroconverted all IBD patients, TNF inhibitor therapy resulted in significantly lower anti-SARS-CoV-2 S Ig antibody levels. Older age also independently showed a negative association with anti-SARS-CoV-2 S Ig antibody levels.²⁴

In our cohort, both biological treatment and combined therapy were associated with lower

serological response compared to AZA and patients without ongoing treatment, however significant differences were not proved during VDZ and UST treatment. Although, the low number of patients in VDZ/UST groups should be enhanced during interpretation of the results. The difference between ongoing treatments was more prominent in participants receiving mRNA-type vaccines. Our post hoc analysis showed similarity of serological response between ustekinumab/vedolizumab and the NONE group which highlights the dissimilarity of different biological agents. However, interpretation of data is limited by low sample sizes in each treatment groups.

Notably, the serological response was higher in the NONE group compared to the HC group. A possible explanation for this phenomenon could be the significantly higher age in the HC group, in accordance with the study mentioned above.²⁴ This relation highlights the potential role of age regarding serological response.

The VARIATON study investigated the effects of mRNA (BNT162b2, CX-024414) and vector (ChAdOx1 nCoV-19, Ad26.CoV2.S) vaccines in IBD patients.²⁵ SARS-CoV-2 S antibody levels were significantly higher after two doses of mRNA vaccines than after administration of vector vaccines. Interestingly IBD itself proved to have a negative impact on anti-spike protein IgG levels. Anti-TNF, anti-IL 12/23 therapy, and Janus kinase (JAK) inhibitors were associated with significantly lower median SARS-CoV-2 levels compared to patients receiving 5-aminosalicylates (5-ASA), immunomodulators, or steroids or no medication. Older age and TNF inhibitor therapy were independent negative confounding factors in the IBD group. No significant difference was observed between TNF inhibitor monotherapy and combination therapy.

Results from a single tertiary IBD center that compared the effects of two doses of mRNA BNT162b2 (Comirnaty; Pfizer-BioNTech, USA), mRNA CX-024414 (Spikevax; Moderna, Cambridge, Massachusetts, USA), or vector ChAdOx1 nCoV-19 (Vaxzevria; AstraZeneca, UK) vaccines on serological response showed that neither biological monotherapy (IFX, ADA, vedolizumab, ustekinumab) nor trough levels were associated with lower SARS-CoV-2 IgG antibody levels. In contrast, variables, such as older age or the combination of biological and immunosuppressive treatment, were identified as negative confounding factors. The lowest antibody levels were found in patients receiving TNF inhibitor and concomitant immunosuppressive treatment (azathioprine/methotrexate). The vector vaccine Vaxzevria was unable to promote seroconversion in 2.2% of IBD patients and induced significantly lower levels of antibodies either in IBD patients or the control group compared to mRNA vaccines.²⁶

Our data showed that mRNA vaccines were superior to non-mRNA types in all groups, excluding vedolizumab treatment. However, the low number of patients receiving VDZ precluded us from drawing significant conclusions. In line with existing international data, our study confirmed the negative effects of older age, combined biological treatment and non-mRNA vaccines on serological response.^{26,27} Based on our results, we therefore highlight the importance of treatment over disease activity on antibody response.

A study by Edelman–Klapper revealed no correlation between anti-TNF drug levels and serological response.²⁴ Our data showed that higher ADA serum levels had a negative effect on anti-SARS-CoV-2 S antibody levels; however, no correlation was observed in cases who received IFX treatment, consistent with the previous study. Our possible hypothesis for this discordance is that the dosage regimen during ADA therapy provides relatively stable drug levels in contrast to IFX, which promotes alternating serum levels. Study protocol did not allow to standardize the time of the sampling of the drug levels due to the real-world setting.

In the current study, we revealed that anti-SARS-CoV-2 S antibody levels persisted for up to 8 weeks after the second dose of the mRNA vaccine. We found no difference between IBD and HC participants during the follow-up period, in contrast to the data published in a few existing studies.^{28,25} Our analysis revealed that vaccination had no significant impact on clinical disease activity based on PGA. Although a statically significant decrease in biochemical activity was observed during follow-up, no clinically significant decrease was noted. A multicenter study by Lev-Tzion *et al.* showed similar exacerbation

rates after vaccination between vaccinated and non-vaccinated IBD patients.²⁹

The strength of the current analysis is our doublecenter, prospective setting with a relatively high number of enrolled patients. Only a few studies have examined the possible correlation between anti-TNF drug levels and serological response. Multivariable analysis has allowed us to review multiple connections. Furthermore, during the study period, Hungary was characterized as one of the countries with highest COVID-19 incidence rates both in Europe and the world, resulting in ingenuous and objective patient selection and enrolment. Notably, only mRNA vaccinations were available in most of the European countries during this period; thus, studies only reported on such vaccines.

The pandemic situation overruled some viewpoints on scientific methodology, resulting in certain limitations in the current study. Testing of serological and therapeutic drug levels in anti-TNF-treated patients was performed at the day of the first vaccination according to the Hungarian immunization protocol, regardless of the treatment cycle. Separated analysis of VDZ, UST, and TOFA groups were not performed due to the low number of patients and potentially misleading results. Biochemical activity was measured by CRP due to its excessive availability; however, fecal calprotectin could provide more accurate data. Almost three times more patients received mRNA vaccines compared to those who received non-mRNA vaccines, which could potentially distort the interpretation of our findings. The proportion of patients enrolled in the study subgroups differed, reflecting the financial protocols in Hungary. Propensity score matching could result in statistical bias during HC selection process.

Based on our double-center, prospective study, anti-SARS-CoV-2 vaccination has considerable efficacy in IBD patients, with mRNA-type vaccines being superior to non-mRNA vaccines. The negative impact of combined biological treatment, especially with high adalimumab drug levels, on serological response to vaccination should be considered with adjustment of vaccination to adalimumab trough level. Midterm durability of vaccination is encouraging; however, more data are needed to expand our existing data in the field of this issue.

Declarations

Ethics approval and consent to participate

The study was approved by the National Institute of Pharmacy and Nutrition according to the Scientific Research Ethics Committee of the Hungarian Medical Research Council's proposal (Registration No. ETT TUKEB IV/861-1/2021/ EKU) and by the Regional and Institutional Human Medical Biological Research Ethics Committee, University of Szeged (approval No.: RKEB 4937). All subjects provided written informed consent prior to study participation. The study was conducted according to the principles of the Declaration of Helsinki.

Consent for publication

TM is the guarantor of the article. All authors have read and agreed to the submitted version of the article.

Author contribution(s)

Tamás Resál: Data curation; Formal analysis; Methodology; Writing – original draft.

Péter Bacsur: Investigation; Methodology; Writing – original draft.

Miklós Horváth: Investigation; Writing – review & editing.

Kata Szántó: Investigation; Writing – review & editing.

Mariann Rutka: Investigation; Writing – review & editing.

Anita Bálint: Investigation; Writing – review & editing.

Anna Fábián: Investigation; Writing – review & editing.

Renáta Bor: Investigation; Writing – review & editing.

Zoltán Szepes: Investigation; Writing – review & editing.

János Fekete: Formal analysis; Investigation; Methodology; Writing – review & editing.

Klaudia Farkas: Conceptualization; Methodology; Writing – review & editing.

Pál Miheller: Conceptualization; Methodology; Writing – review & editing.

Tamás Molnár: Conceptualization; Methodology; Writing – review & editing.

Acknowledgements

Writing support was provided by Enago and funded by Klebelsberg Library, University of Szeged. The authors are grateful to Csilla Tóth-Káli, Gabriella Pócsik, Brigitta Csorba, and Krisztina Ördög for study support (*Department of Medicine, University of Szeged*).

Funding

The authors disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This work was supported by research grants from the National Research, Development and Innovation Office (Grant ID: 143549, 125377, 129266, 139269 and 134863) and the New National Excellence Programme of the Ministry of Human Capacities (UNKP-22-3-SZTE-233 to BP, UNKP-22-4 -SZTE-293 to BR, UNKP-22-3-SZTE-278 to RT, UNKP-22-4-SZTE-296 to FA) and by a J János Bolyai Research Grant nos Bolyai Research Grant (BO/00723/22) and a Géza Heténvi Research Grant (to FK, RM and BA) from the Faculty of Medicine, University of Szeged.

Competing interests

Tamás Molnár has received speaker's honoraria from MSD, AbbVie, Egis, Goodwill Pharma, Takeda, Pfizer, Janssen, Sandoz, MundiPharma, Phytotec, Roche, Fresenius, and Teva. Klaudia Farkas has received speaker's honoraria from AbbVie, Janssen, Ferring, Takeda, and Goodwill Pharma. PB, MH, KSZ, MR, AB, AF, RB, ZSZ, JF, and PM have no conflict of interest to disclose.

Availability of data and materials

The data underlying this article will be shared on reasonable request to the corresponding author. The current article, including related data and figures, has not been previously published and is not under consideration elsewhere.

ORCID iDs

Tamás Resál 3842-9094	ÍD	https://orcid.org/0000-0002-
Péter Bacsur 8534-0068	D	https://orcid.org/0000-0002-
Kata Szántó 0749-5061	D	https://orcid.org/0000-0003-

0824-7476

Anna Fábián D https://orcid.org/0000-0002-

Renáta Bor https://orcid.org/0000-0001-9393-5240

Supplemental material

Supplemental material for this article is available online

References

- 1. WHO Director-General's opening remarks at the media briefing on COVID-19-11 March 2020, https://www.who.int/dg/speeches/detail/whodirector-general-s-opening-remarks-at-the-mediabriefing-on-covid-19-11-march-2020
- 2. Derikx L, Lantinga MA, de Jong DJ, et al. Clinical outcomes of Covid-19 in patients with inflammatory bowel disease: a Nationwide Cohort Study. J Crohns Colitis 2021; 15: 529-39.
- 3. Rahier JF, Magro F, Abreu C, et al. Second European evidence-based consensus on the prevention, diagnosis and management of opportunistic infections in inflammatory bowel disease. 7 Crohns Colitis 2014; 8: 443-468.
- 4. Brenner EJ, Ungaro RC, Gearry RB, et al. Corticosteroids, but not TNF antagonists, are associated with adverse COVID-19 outcomes in patients with inflammatory bowel diseases: results From an International Registry. Gastroenterology 2020; 159: 481-91.e3.
- 5. Wisniewski A, Kirchgesner J, Seksik P, et al. Increased incidence of systemic serious viral infections in patients with inflammatory bowel disease associates with active disease and use of thiopurines. United Eur Gastroenterol 7 2020; 8: 303-313.
- 6. Singh AK, Jena A, Kumar MP, et al. Risk and outcomes of coronavirus disease in patients with inflammatory bowel disease: a systematic review and meta-analysis. United Eur Gastroenterol J 2021; 9: 159-176.
- 7. Wellens J, Colombel JF, Satsangi JJ, et al. SARS-CoV-2 Vaccination in IBD: Past Lessons, Current Evidence, and Future Challenges. 7 Crohns Colitis 2021; 15: 1376-1386.
- 8. Prentice RE, Rentsch C, Al-Ani AH, et al. SARS-CoV-2 vaccination in patients with inflammatory bowel disease. Gastro Hep 2021; 3: 212-228.

- 9. Lamb CA, Kennedy NA, Raine T, et al. British Society of Gastroenterology consensus guidelines on the management of inflammatory bowel disease in adults. Gut 2019; 68: s1-s106.
- 10. Siegel CA, Melmed GY, McGovern DP, et al. SARS-CoV-2 vaccination for patients with inflammatory bowel diseases: recommendations from an international consensus meeting. Gut 2021; 70: 635-640.
- 11. Bhurwal A, Mutneja H, Bansal V, et al. Effectiveness and safety of SARS-CoV-2 vaccine in Inflammatory Bowel Disease patients: a systematic review, meta-analysis and meta-regression. Aliment Pharmacol Ther 2022; 55: 1244-1264.
- 12. Jena A, James D, Singh AK, et al. Effectiveness and durability of COVID-19 vaccination in 9447 patients with IBD: a systematic review and meta-analysis. Clin Gastroenterol Hepatol 2022; 20: 1456-1479.e18.
- 13. von Elm E, Altman DG, Egger M, et al. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. 7 Clin Epidemiol 2008; 61: 344-349.
- 14. Merkely B, Szabó AJ, Kosztin A, et al. Novel coronavirus epidemic in the Hungarian population, a cross-sectional nationwide survey to support the exit policy in Hungary. Geroscience 2020; 42: 1063–1074.
- 15. Silverberg MS, Satsangi J, Ahmad T, et al. Toward an integrated clinical, molecular and serological classification of inflammatory bowel disease: Report of a Working Party of the 2005 Montreal World Congress of Gastroenterology. Can 7 Gastroenterol 2005;19(Suppl. A): 5e36.
- 16. Best WR, Becktel JM, Singleton JW, et al. Development of a Crohn's disease activity index. National Cooperative Crohn's Disease Study. Gastroenterology 1976; 70: 439-44.
- 17. Schroeder KW, Tremaine WJ and Ilstrup DM. Coated oral 5-aminosalicylic acid therapy for mildly to moderately active ulcerative colitis. A randomized study. N Engl J Med 1987; 317: 1625-1629.
- 18. Diederen K, Gerritsma JJ, Koot BGP, et al. Do Children and adolescents with inflammatory bowel disease complete clinical disease indices similar to physicians? J Pediatr Gastroenterol Nutr 2018; 66: 410-416.
- 19. Singh S, Murad MH, Fumery M, et al. First- and second-line pharmacotherapies for patients with moderate to severely active ulcerative colitis: an

updated network meta-analysis. *Clin Gastroenterol Hepatol* 2020; 18: 2179–2191.e6.

- Torres J, Bonovas S, Doherty G, et al. ECCO Guidelines on Therapeutics in Crohn's Disease: Medical Treatment. J Crohns Colitis 2020; 14: 4–22.
- Mathur A, Sahu S, Rai S, *et al.* Serological response to vaccination against coronavirus disease-19 in patients with inflammatory bowel disease. *Indian J Gastroenterol* 2023; 42: 64–69.
- Kennedy NA, Lin S, Goodhand JR, et al. Infliximab is associated with attenuated immunogenicity to BNT162b2 and ChAdOx1 nCoV-19 SARS-CoV-2 vaccines in patients with IBD. Gut 2021; 70: 1884–1893.
- Chanchlani N, Lin S, Chee D, et al. Adalimumab and infliximab Impair SARS-CoV-2 antibody responses: results from a therapeutic drug monitoring study in 11 422 biologic-treated patients. J Crohns Colitis 2022; 16: 389–397.
- Edelman-Klapper H, Zittan E, Bar-Gil Shitrit A, et al. Lower serologic response to COVID-19 mRNA vaccine in patients with inflammatory bowel diseases treated with anti-TNFα. *Gastroenterology* 2022; 162: 454–467.
- 25. Doherty J, O Morain N, Stack R, *et al.* Reduced serological response to

COVID-19 vaccines in patients with IBD is further diminished by TNF inhibitor therapy; Early results of the VARIATION study (VAriability in Response in IBD against SARS-COV-2 ImmunisatiON). *J Crohns Colitis* 2022; 16: 1354–1362.

- Cerna K, Duricova D, Lukas M, et al. Anti-SARS-CoV-2 vaccination and antibody response in patients with inflammatory bowel disease on immune-modifying therapy: prospective single-tertiary study. *Inflamm Bowel Dis* 2022; 28(10):1506–1512.
- Alexander JL, Kennedy NA, Ibraheim H, et al. COVID-19 vaccine-induced antibody responses in immunosuppressed patients with inflammatory bowel disease (VIP): a multicentre, prospective, case-control study. *Lancet Gastroenterol Hepatol* 2022; 7: 342–352.
- 28. Vollenberg R, Tepasse PR, Kühn JE, *et al.* Humoral immune response in IBD patients three and six months after vaccination with the SARS-CoV-2 mRNA vaccines mRNA-1273 and BNT162b2. *Biomedicines* 2022; 10: 171.
- 29. Lev-Tzion R, Focht G, Lujan R, *et al.* COVID-19 vaccine is effective in inflammatory bowel disease patients and is not associated with disease exacerbation. *Clin Gastroenterol Hepatol* 2022; 20: e1263–e1282.

Visit Sage journals online

journals.sagepub.com/

Sage journals

home/tag